Author's response to reviews

Title: Immunity status of adults and children against poliomyelitis virus type 1 strains CHAT and Sabin (LSc-2ab) in Germany

Authors:

Maren Eggers (eggers@labor-unders.de)
Elena Terletskiaia-Ladwig (ladwig@labor-unders.de)
Holger F Rabenau (Rabenau@em.uni-frankfurt.de)
Hans W Doerr (H.W.Doerr@em.uni-frankfurt.de)
Sabine Diedrich (DiedrichS@rki.de)
Gisela Enders (enders@labor-unders.de)
Martin Enders (menders@labor-unders.de)

Version: 4 Date: 3 November 2010

Author's response to reviews: see over
Dear Ms Rajabi,

Thank you for accepting our manuscript for publication.

Answers to your questions and remarks are listed below:

- Please can you put all of the author's email addresses on the title page.
  All author’s e-mails are put on the title page

- Please can you name the ethical committee that approved your study.
  Ethical approval was not required. With respect to this, we completed our manuscript
  with more detailed information. The study was carried out in compliance with the
  Helsinki Declaration. None of the samples was collected for study purposes.
  Discarded sera from our routine diagnostics were de-identified. Sera from laboratory
  workers were coded. All samples were tested in an anonymized fashion. Please see
  page 7 line 137 to 144 and page 8 line 155 to 158.

- Please include a 'Competing interests' section between the Conclusions and
  Authors' contributions. If there are none to declare, please write 'The authors declare
  that they have no competing interests'.
  A 'Competing interests' section is included.

- In the past five years have you received reimbursements, fees, funding, or salary
  from an organization that may in any way gain or lose financially from the publication
  of this manuscript, either now or in the future? Is such an organization financing this
  manuscript (including the article-processing charge)? If so, please specify.
  No

- Do you hold any stocks or shares in an organization that may in any way gain or
  lose financially from the publication of this manuscript, either now or in the future? If
  so, please specify.
  No

- Do you hold or are you currently applying for any patents relating to the content of
  the manuscript? Have you received reimbursements, fees, funding, or salary from an
  organization that holds or has applied for patents relating to the content of the
  manuscript? If so, please specify.
  No

- Do you have any other financial competing interests? If so, please specify.
  No

The changes are highlighted using Microsoft Office Word and the conformity of
manuscript to the journal style was checked.

Best regards,
Maren Eggers